Author's response to reviews

Title: Successfully tested but not enrolled in the care for human immunodeficiency virus co-infected tuberculosis patients: an observational cohort study at primary care centers in a rural district, Zambia

Authors:

Shinsuke Miyano (s-miyano@it.ncgm.go.jp)
Samba Muvuma (smuvuma@yahoo.com)
Naoko Ishikawa (n-ishikawa@it.ncgm.go.jp)
Hiroyoshi Endo (endo@research.twmu.ac.jp)
Charles Msiska (Myekha@yahoo.com)
Gardner Syakantu (gsyakantu@yahoo.com)

Version: 2 Date: 14 February 2013

Author's response to reviews: see over